

LJMU Research Online

Rathmann, H, Perretti, S, Porcu, V, Hanihara, T, Scott, GR, Irish, JD, Reyes-Centeno, H, Ghirotto, S and Harvati, K

Inferring human neutral genetic variation from craniodental phenotypes

http://researchonline.ljmu.ac.uk/id/eprint/20272/

Article

Citation (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Rathmann, H, Perretti, S, Porcu, V, Hanihara, T, Scott, GR, Irish, JD, Reyes-Centeno, H, Ghirotto, S and Harvati, K (2023) Inferring human neutral genetic variation from craniodental phenotypes. PNAS Nexus, 2 (7).

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact researchonline@ljmu.ac.uk

http://researchonline.ljmu.ac.uk/



https://doi.org/10.1093/pnasnexus/pgad217 Advance access publication 3 July 2023 Research Report

Inferring human neutral genetic variation from craniodental phenotypes

Hannes Rathmann (D^{a,b,*}, Silvia Perretti (D^c, Valentina Porcu^c, Tsunehiko Hanihara^d, G. Richard Scott (D^e, Joel D. Irish (D^{f,g}, Hugo Reyes-Centeno (D^{h,i,j}, Silvia Ghirotto (D^c and Katerina Harvati (D^{a,b,j})

^aSenckenberg Centre for Human Evolution and Palaeoenvironment, University of Tübingen, Tübingen 72070, Germany

^bPaleoanthropology, Institute for Archaeological Sciences, Department of Geosciences, University of Tübingen, Tübingen 72070, Germany

^cDepartment of Life Sciences and Biotechnology, University of Ferrara, Ferrara 44121, Italy

^dDepartment of Anatomy, Kitasato University School of Medicine, Sagamihara 252-0374, Japan

^eDepartment of Anthropology, University of Nevada, Reno, NV 89557, USA

^fResearch Centre in Evolutionary Anthropology and Palaeoecology, School of Biological and Environmental Sciences, Liverpool John Moores University, Liverpool L3 3AF, UK

^gThe Centre for the Exploration of the Deep Human Journey, University of the Witwatersrand, Johannesburg WITS 2050, South Africa

^hDepartment of Anthropology, University of Kentucky, Lexington, KY 40506, USA

ⁱWilliam S. Webb Museum of Anthropology, University of Kentucky, Lexington, KY 40504, USA

^jDFG Center for Advanced Studies 'Words, Bones, Genes, Tools', University of Tübingen, Tübingen 72070, Germany

*To whom correspondence should be addressed: Email: hannes.rathmann@uni-tuebingen.de

Edited By: George Milner

Abstract

There is a growing consensus that global patterns of modern human cranial and dental variation are shaped largely by neutral evolutionary processes, suggesting that craniodental features can be used as reliable proxies for inferring population structure and history in bioarchaeological, forensic, and paleoanthropological contexts. However, there is disagreement on whether certain types of data preserve a neutral signature to a greater degree than others. Here, we address this unresolved question and systematically test the relative neutrality of four standard metric and nonmetric craniodental data types employing an extensive computational genotype-phenotype comparison across modern populations from around the world. Our computation draws on the largest existing data sets currently available, while accounting for geographically structured environmental variation, population. Our results reveal that the four data types differentially capture neutral genomic variation inherent to a neutral model of evolution. Our results and cranial metric data, followed by cranial nonmetric and dental metric data. Importantly, we demonstrate that combining all four data types together maximizes the neutral genetic signal compared with using them separately, even with a limited number of phenotypic variables. We hypothesize that this reflects a lower level of genetic integration through pleiotropy between, compared to within, the four data types, effectively forming four different modules associated with relatively independent sets of loci. Therefore, we recommend that future craniodental investigations adopt holistic combined data approaches, allowing for more robust inferences about underlying neutral genetic variation.

Keywords: cranium, dentition, neutral evolution, genetic variation

Significance Statement

Craniodental features are routinely used in bioarchaeology, forensics, and paleoanthropology to infer genetic relatedness across human remains. However, it is unclear whether certain data types preserve neutral evolutionary signals to a greater degree than others. Here, we test the relative utility of four standard metric and nonmetric data types, employing an extensive computational genotypephenotype comparison across worldwide modern populations. Our results reveal that the four data types capture different amounts of neutral genomic variation, with dental nonmetrics and cranial metrics showing the highest signals and dental metrics displaying the lowest. Importantly, combining different data types maximizes genotypic coverage over different loci compared with using them separately. Therefore, we recommend prioritizing combined data sets for more accurate craniodental inferences in future research.

Introduction

Human skeletal morphology is highly diverse and varies among individuals and populations across the globe. This pattern was shaped by the complex interplay of neutral evolutionary processes (i.e. selectively neutral mutations, random genetic drift, and gene flow) and nonneutral forces related to local adaptation

Competing Interest: The authors declare no competing interest.

Received: April 12, 2023. Revised: June 14, 2023. Accepted: June 20, 2023



© The Author(s) 2023. Published by Oxford University Press on behalf of National Academy of Sciences. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

and developmental plasticity in response to environmental and cultural stimuli (1–3). Different parts of the skeleton (such as the cranium, mandible, teeth, pelvis, long bones, hands, and feet) have been shown to preserve neutral and nonneutral signatures to different degrees (4–13). Overall, however, there is wide consensus that cranial and dental morphology as a whole evolved for a large part under neutrality and, thus, can be used as a proxy for reconstructing population structure and history (14–20). This is relevant for the study of human skeletal remains from archaeological and forensic contexts, where DNA analyses are often constrained due to poor molecular preservation, particularly in the deep fossil record and in warmer climates, or when destructive DNA sampling of fragile and rare specimens is not possible.

Morphological investigations based on craniodental features typically focus on either quantitative (hereafter, metric) or qualitative (hereafter, nonmetric) data to characterize the overall geometry of study specimens. Cranial metric data collection is performed by defining a set of homologous anatomical landmarks located on the skull and by measuring either linear dimensions between them (21, 22) or the relative position of landmark and semilandmark coordinates in two or three dimensions (23-26). Dental metric data collection is performed in a similar fashion, usually by measuring linear lengths, widths, or diagonal dimensions at the tooth crown or at the cement-enamel junction (27, 28) or semilandmark-based crown outlines (29, 30). Cranial nonmetric trait data collection is performed by visually scoring minor discontinuous variants, such as extra-sutural ossicles, proliferative ossifications including bridges or spurs, or variation in foramina number and location, for example (31-33). Similarly, dental nonmetric trait data collection is performed by observing the number of cusps and roots, or the pattern of fissures, ridges, and grooves on tooth crowns (34–37).

Despite the popularity of all four craniodental data types in population structure and history studies, it remains poorly understood whether some preserve neutral genomic signatures to a greater degree than others. This is problematic because investigations based on different craniodental data types may arrive at markedly disparate conclusions. For example, some researchers have suggested that teeth are a "safe box" of the genetic code, much more than any other skeletal element, because they form relatively early during ontogeny and their morphology remains unchanged after full formation, making teeth less affected by external stimuli (38). Some have also hypothesized that metric data are more useful than nonmetric traits, because measurements can be collected in a more objective and consistent manner, whereas visual scoring of nonmetric traits can be subjective and prone to observer error (39, 40). A vast body of literature also suggest varying levels of heritability among the different craniodental data types, with disparate amounts of genetic integration through pleiotropy, indicating that some types of data contain more independent genomic information than others (41-48). Several studies also point out that more holistic approaches combining different craniodental data types in a single analysis capture more phenotypic and thus genomic variation, compared with using them separately (49-51). Lastly, it has been proposed that there are not only differences in neutrality between the craniodental data types, but also differences within a given data type. That is, some bones, single trait expressions, or functional and developmental modules conserve a stronger evolutionary neutral signal than other, more labile, regions (10, 14, 17, 20).

A standard approach for quantifying the utility of a given craniodental data type in capturing a neutral genomic signature is to estimate phenotypic distances among worldwide modern human populations, on the one hand, and to compare them to neutral genomic distances estimated among the same or closely matched set of populations on the other (1, 2, 52). These analyses, hereafter termed $D_{\rm P}$ - $D_{\rm G}$ comparisons, have been extensively performed for cranial metric data (14, 16, 17, 19, 20, 51, 53, 54), dental metric data (18, 55), cranial nonmetric trait data (51, 56, 57), and dental nonmetric trait data (10, 15, 18, 58). However, the estimated levels of neutrality of the different craniodental data types reported in previous $D_{\rm P}$ - $D_{\rm G}$ studies are not directly comparable, since different populations have been sampled and diverse methodological approaches for calculating between-population distances have been employed at different geospatial scales (54).

To date, only few D_P-D_G studies have attempted to systematically co-analyze the relative neutrality of different craniodental data types in a single analytical framework, thus, allowing for comparability (18, 51, 56). Those investigations found contradicting results, reporting either similar degrees of neutrality for different data types (18) or that they were differentially associated with genomic markers (51, 56). However, those previous studies were constrained by several factors. First, they were limited to either cranial (51, 56) or dental (18) data and none compared all four craniodental data types together. Second, none of the previous studies assessed the utility of a mixed-type data set combining metric and nonmetric traits in a single analysis. Third, none of these studies accounted for geographically structured environmental variation that can affect phenotypic and genomic variation (12). Fourth, all used rather limited sets of matched populations with varying and sometimes small sample sizes without accounting for variation introduced by sampling uncertainty. Fifth, all studies compared craniodental data types with unequal numbers of variables, which leads to biased results since phenotypic distances based on many variables are more robust than those based on only a few (10, 59). Sixth and finally, all previous studies compared phenotypic distances to a single point estimate of genetic distance, which takes all sampled genomic loci into consideration; instead, phenotypic distances should be compared with multiple equally plausible neutral genetic distances by randomly sampling genomic loci in order to account for stochastic variation inherent to a neutral model of evolution (10, 52, 60).

In this study, we address these research gaps by using a global $D_{\rm P}$ - $D_{\rm G}$ framework in which we jointly investigate the relative neutrality of the four different craniodental data types, plus a mixed-type data set combining all four types of data together. Our extensive computations draw on the largest existing genomic and phenotypic databases currently available, while accounting for geographically structured environmental variation, population sampling uncertainty, disparate numbers of phenotypic variables, and stochastic variation inherent to a neutral model of evolution.

Results

Mining large existing databases, we matched five different genomic and phenotypic data types for 26 modern human population samples from around the world, namely: (i) 8,821 single nucleotide polymorphisms (SNPs), (ii) 37 cranial metrics (in the form of linear dimensions, arcs, chords, and subtenses), (iii) 28 dental metrics (in the form of mesiodistal and buccolingual crown diameters), (iv) 24 cranial nonmetric traits, and (v) 25 dental nonmetric traits (Fig. 1A and SI Appendix, Table S1). We then estimated pairwise between-population genetic distances using Weir-Cockerham's F_{ST} derived from the SNP data, which served as a benchmark to evaluate neutral expectations (Data Set S1). Next,



Fig. 1. Geographic location and genomic and phenotypic relationships of worldwide modern human populations. (A) World map showing the locations of 26 populations sampled for matched genomic data (SNPs) and phenotypic data (cranial and dental metrics and nonmetric traits). Points are approximate geographic coordinates of the genomic samples. 3D NMDS plots of between-population distances, calculated separately from six different data types: (B) SNPs; (C) cranial metrics; (D) dental metrics; (E) cranial nonmetric traits; (F) dental nonmetric traits; and (G) combined craniodental data.

we estimated pairwise between-population phenotypic distances using Mahalanobis' D^2 , generated separately from the cranial metrics, dental metrics, cranial nonmetric traits, dental nonmetric traits, and the combined craniodental data (Data Sets S2–S6). We then subjected the F_{ST} and D^2 distances to Kruskal's nonmetric multidimensional scaling (NMDS) to visualize the matrices in a decomposed three-dimensional (3D) coordinate space, where a spatial grouping of populations indicates close affinity, and vice versa (Fig. 1B–G). The MDS stress level for the F_{ST} matrix was 0.0407, and the stress levels for the cranial metric, dental metric, cranial nonmetric traits, dental nonmetric traits, and combined craniodental data D^2 matrices were 0.0539, 0.1185, 0.1392, 0.0683, and 0.0615, respectively. All stress levels are below the acceptable threshold of 0.15, indicating that 3D captures the overall

among-population variation well. To spatially orient the D^2 distance configurations similar to the F_{ST} distance configuration, we subjected the decomposed D^2 coordinates to Procrustes superimposition to scale and rotate them to maximum similarity with the decomposed F_{ST} coordinates by minimizing the overall sum of squared differences among populations. All 3D NMDS plots show major continental clusters of populations. The clusters appear most markedly geographically structured in the SNP data (Fig. 1B) and to a similar degree in the combined craniodental data (Fig. 1G), whereas clusters in the dental metric data appear least structured geographically (Fig. 1D).

To formally quantify the neutral signals preserved in a given phenotypic data type, we conducted partial correlation tests to measure the degree of congruence between D^2 and F_{ST} , while controlling for the effects of geographically structured environmental variation on phenotypic and genomic variation (12). Computationally, the partial correlation test design calculates the correlation of the residuals from the independent regressions $D^2 \sim C$ and $F_{ST} \sim C$, whereby C describes climatic differences among sampled population environments (Data Set S7). The resulting partial correlation value r was treated as a neutrality estimate, with an r value close to 1 indicating a higher degree of neutrality, whereas an r value near to 0 indicates a lower degree. We obtained the highest r value for the combined craniodental data (r = 0.684), followed by cranial metrics (r = 0.618), dental nonmetric traits (r = 0.592), cranial nonmetric traits (r = 0.390), and dental metrics (r = 0.223). Similar patterns were observed when comparing D^2 to F_{ST} , while controlling for geographic distances (G) (Data set S8), albeit with slightly lower overall r values (SI Appendix, Table S2). However, due to variations in sample sizes between the matched phenotypic and genomic data sets (SI Appendix, Table S1), the D^2 and F_{ST} distances are statistically biased, and in consequence, the neutrality estimate r. Therefore, to explore the effect of population sampling uncertainty, we employed a resampling procedure whereby we calculated the neutrality estimator 1,000 times, each time leaving out a randomly selected population in the phenotypic and genomic data sets and a randomly selected individual in each remaining population. We then reported the median of the resulting distribution of r values and constructed an interpercentile range accounting for 95% of the spread. The results are summarized in Table 1 and visualized in Fig. 2A using violin plots. Overall, the highest distribution of r values was again attained for the combined craniodental data, followed by cranial metrics, dental nonmetric traits, cranial nonmetric traits, and dental metrics. To statistically corroborate this finding, we conducted repeated-measures t-tests among pairs of distributions (SI Appendix, Table S3) and found significant differences in the levels of neutral signals preserved in each craniodental data type (P < 0.001).

The five phenotypic data sets in our analysis comprise unequal numbers of variables (SI Appendix, Table S1), namely: 37 cranial metric variables; 24 cranial nonmetric trait variables; 28 dental metric variables; 25 dental nonmetric trait variables; and the combined craniodental data set comprises a summed up total of 114 variables. This imbalance hampers a direct comparison across data types, given that phenotypic analyses based on many variables are more robust than those based on only a few (10, 59). Therefore, to create equally sized numbers of variables across all five phenotypic data sets, we calculated the neutrality estimator r for a given phenotypic data type 1,000 times, each time randomly undersampling the number of variables down to 24. This corresponds to the number of variables in the cranial nonmetric trait data set, comprising the fewest variables among all data sets compared. Phenotypic variable sampling bias correction was performed together with population sampling bias correction, to explore the combined effect of these two analytical refinements. On average, the resulting distributions of r values for the five phenotypic data types exhibit a similar relative ordering to the population sampling bias corrected r values alone, with the only difference that dental nonmetric traits show a higher preservation of neutral genomic signatures compared with cranial metrics (Table 1, Fig. 2B). Pairwise repeated-measures t-tests (SI Appendix, Table S4) confirmed that the neutral signals preserved in each craniodental data type significantly differ from one another (P < 0.001).

Under a neutral model of evolution, the F_{ST} distance matrix, used as a benchmark for our comparisons, is just one of multiple equally plausible neutral genetic outcomes produced by stochastic variation (10, 52, 60). To account for this stochasticity, we calculated F_{ST} and thus the neutrality estimator r for a given phenotypic data type 1,000 times, each time randomly undersampling the number of SNP loci down to the same number of phenotypic variables, namely 24. This sampling strategy is consistent with population and quantitative genetics theory, where a heritable, additive, and selectively neutral phenotype is approximately as informative about population differentiation as a single neutral genomic locus, regardless of how many loci influence the phenotype (61, 62). Loci undersampling was performed in conjunction with population sampling bias correction and phenotypic variable sampling bias correction, to investigate the combined effect of these three analytical refinements. On average, the resulting distributions of r values exhibit a similar relative ordering to those correcting for population and phenotypic variable sampling bias combined (Table 1, Fig. 2C). Pairwise repeated-measures t-tests (SI Appendix, Table S5) showed that all neutral signals differ significantly (P < 0.001).

We note that overall our phenotypic data sets exhibit imbalanced distributions of sexes, with more males represented than females. This bias can be problematic as sexual dimorphism in shape, size, and trait expression may introduce variation unrelated to neutral genomic variation. Although we implemented data preprocessing steps to correct for sexual dimorphism (see Materials and Methods), we conducted two additional analyses focusing on males only. One analysis utilized size-corrected metric data (SI Appendix, Table S6), while the other did not apply size correction to the metric data (SI Appendix, Table S7). The results from the male subsets generally follow the same pattern as those of the complete data set, albeit with slightly wider *r* value intervals when controlling for population sampling uncertainty, which is expected given the overall smaller sample size.

Lastly, we acknowledge that our results may be affected by small sample sizes for certain data types, particularly dental metrics and SNPs, which were represented by only a few individuals in some populations (SI Appendix, Table S1). Although our analysis accounts for sampling bias (as described above), we conducted a more cautious analysis focusing on a subset of 16 out of the 26 populations with phenotypic and genomic sample sizes of $n \ge 10$ (SI Appendix, Table S8). The obtained results show patterns that generally align with those observed in the full data set, although with slightly higher overall r values. The only distinction lies in higher r values observed for cranial metrics compared with dental nonmetric traits. Therefore, in order to reconcile the findings of the full 26-population data set and the 16-population subset, we consider dental nonmetric traits and cranial metrics equally suitable for tracking neutral signatures until further samples become available for study.

Table 1. Neutrality estimates for five craniodental data types, calculated as partial Pearson correlation (r) between phenotypic (D^2) and neutral genetic (F_{ST}) distances across 26 modern human population samples, controlling for climate distances.

Craniodental data type	Population sampling bias correction ^a	Population and phenotype sampling bias correction ^b	Population, phenotype, and loci sampling bias correction ^c
Cranial metrics	0.610 (0.553–0.682)	0.573 (0.498–0.640)	0.455 (0.276–0.610)
Cranial nonmetric traits	0.380 (0.346–0.430)	0.380 (0.346-0.430)	0.309 (0.138-0.464)
Dental metrics	0.194 (0.118–0.281)	0.190 (0.085–0.301)	0.150 (-0.043-0.338)
Dental nonmetric traits	0.591 (0.534–0.633)	0.586 (0.528–0.642)	0.476 (0.261–0.652)
Craniodental combined	0.672 (0.627–0.743)	0.644 (0.568–0.716)	0.522 (0.349–0.668)

^aMedian (and 95% range) of 1,000 iteratively generated *r* values, each iteration leaving out a randomly selected population in the phenotypic and genomic data sets and a randomly selected individual in each remaining population.

^bMedian (and 95% range) of 1,000 iteratively generated r values, each iteration randomly undersampling the number of phenotypic variables, combined with population sampling bias correction. ^cMedian (and 95% range) of 1,000 iteratively generated r values, each iteration randomly undersampling the number of loci, combined with population and

^cMedian (and 95% range) of 1,000 iteratively generated *r* values, each iteration randomly undersampling the number of loci, combined with population and phenotype sampling bias correction.

Discussion

To our knowledge, this study is the first to systematically co-analyze the relative utility of four widely used standard craniodental phenotypic data types in capturing neutral genomic variation, namely (i) cranial metrics, (ii) dental metrics, (iii) cranial nonmetric traits, and (iv) dental nonmetric traits, plus (v) a mixedtype data set combining all four data types together. We performed a comprehensive D_P-D_G comparison across 26 worldwide populations, drawing on the largest existing phenotypic and genomic data sets currently available, and incorporating a range of analytical refinements commonly neglected in previous D_P-D_G studies. In doing so, we demonstrated the importance of accounting for sampling uncertainty and showed that r neutrality estimates can vary substantially based on the composition of population samples and numbers of specimens included, even with large data sets as employed here. This is, for example, markedly expressed by the dental metric data in the full 26-population data set, with a point estimate of r = 0.223, which widens to a 95% range of r = 0.118-0.281 when accounting for sampling bias. We further demonstrated the importance of accounting for unevenly sized numbers of phenotypic variables when comparing relative levels of neutrality across phenotypic data sets. Specifically, in the full 26-population data set, cranial metrics exhibited higher levels of neutrality compared with dental nonmetric traits when no correction was applied, but this pattern reversed when the number of phenotypic variables was equalized across data sets through random undersampling. This result is in agreement with previous research finding that the validity of cranial metric and dental nonmetric trait distances in reflecting neutral expectations is contingent upon the number of variables employed (10, 59). On a related note, our undersampling procedure also takes into account the practical limitations of working with skeletal remains, particularly in bioarchaeological or fossil contexts, where craniodental data are often highly fragmented, and where researchers must work with random subsets of variables. Lastly, we demonstrated the importance of accounting for stochastic variation inherent to a neutral model of evolution by randomly undersampling the SNP loci to match the number of phenotypic variables. This resulted in r neutrality estimate distributions with much wider ranges, and for the dental metric data, the 95% range was found to be r = -0.043-0.338 (in the full 26-population data set) and r = 0.067 - 0.612 (in the 16-population subset), with the lower bounds near zero implying nonneutral evolutionary forces. This finding therefore calls into question the validity of dental metrics as a proxy for neutral genomic markers.

Inspecting the four craniodental data types separately, our results clearly show that they are differentially associated with neutral genomic variation after accounting for population sampling uncertainty, disparate numbers of phenotypic variables, and stochastic variation inherent to a neutral model of evolution. In testing for neutrality, our estimates reveal that, overall, dental nonmetric traits and cranial metrics performed best, followed at some distance by cranial nonmetric traits, whereas dental metrics performed relatively poorly. Interestingly, these estimates do not relate to the suggested general divide in utility between cranial versus dental features, with the latter proposed to be less affected by external environmental stimuli (38), and nonmetric versus metric data, with the latter suggested to be less prone to observer error (39). Instead, our estimates are in agreement with previous quantitative genetic studies of pleiotropy in humans (or in nonhuman primates when studies on humans are not yet available), finding that the amount of independent genetic information in dental metrics (41) and cranial nonmetric traits (45) is low, compared with the amount of independent genetic information in cranial metrics (44) and dental nonmetric traits (42, 43). The relatively poor performance of dental metrics contrasts with what was proposed in a previous study using a methodological $D_{\rm P}$ - $D_{\rm G}$ framework similar to ours (18), which found that dental metrics and nonmetric traits are both comparably well-suited in tracking neutral genomic variation. The present study expands and improves upon the $D_{\rm P}$ - $D_{\rm G}$ investigation by Rathmann et al. (18) in several respects. Among the most important are a more comprehensive dental nonmetric trait data set for comparison (25 versus 15 traits) and a larger set of globally distributed matched population samples (26 versus 19 populations).

Perhaps one of the most interesting findings of our study is that phenotypic inferences of neutral genomic variation are most accurate when based on a combined mixed-type data set, compared with using the four different data types separately. This result is in agreement with previous studies reporting that phenotypic inferences about genomic affinities are best served when multiple lines of evidence are jointly investigated (49-51). This is also expected, given that the number of variables in the mixed-type data set is many times higher than in the four different data sets separately, leading to a richer knowledge of phenotypic and thus genomic variation (10, 59). Interestingly though, when equalizing the numbers of phenotypic variables across all data sets via undersampling, the mixed-type data still performed best. One possible explanation for this result could be that genetic integration through pleiotropy between the four data types is lower than genetic integration within the four data types, effectively forming four different modules regulated by different sets of loci that are relatively independent from each other (63-65). In this situation, even when just a few phenotypic variables per data type would



Fig. 2. Violin plots showing neutrality estimates for five craniodental data types, calculated as partial Pearson correlation (r) between phenotypic (D^2) and neutral genetic (F_{ST}) distances across 26 modern human population samples, controlling for climate distances. Box plots are superimposed to show median values (black solid line) and interquartile ranges (boxes). (A) Distribution of 1,000 iteratively generated r values, each iteration leaving out a randomly selected population in the phenotypic and genomic data sets and a randomly selected individual in each remaining population. (B) Distribution of 1,000 iteratively generated r values, each iteration randomly undersampling the number of phenotypic variables, combined with population sampling bias correction. (C) Distribution of 1,000 iteratively generated r values, each iteratively generated r values, each iteration and phenotype sampling bias correction.

contribute to the mixed-type data, more underlying genomic variation from different loci would still be captured than using the full phenotypic variable battery of one of the four data types individually. This hypothesis could be tested with a quantitative genetic analysis of pleiotropy in a modern human population with known pedigree structure sampled for all four cranial and dental metric and nonmetric trait data types, which to our knowledge has not been performed so far and could thus lead to exciting new research directions.

We note that the reported r neutrality estimates for the different craniodental data types must be considered minimum values as they are biased toward zero. This is because we compared matched but unpaired population samples, with phenotypic and genomic data coming from different individuals; however, phenotypic and genomic distances among unpaired samples have a reduced degree of congruence, given that within-population variation is high compared with between-population variation (66). Nevertheless, comparing unpaired samples is a standard procedure in global scale D_P - D_C analyses (7, 10, 14–20, 51, 55), and our applied analytical correction for sampling bias (i.e. both population and specimen resampling of the phenotypic and genomic data) may account for at least some of this uncertainty. Moreover, although our large craniodental data sets comprise the most widely used metric and nonmetric trait variables in bioanthropological research, they could be complemented with additional standard and nonstandard variables proposed to be informative (67-70). Similarly, the metric portion of our data sets, consisting of linear dimensions, arcs, cords, and subtenses, could be replaced with 3D coordinate data that better retain the geometry of the studied specimens than caliper-based measurements. Interestingly,

though not fully comparable, previous D_P-D_G analyses based on craniodental 3D data reported neutrality levels similar to those reported here (14, 17, 19, 20, 55), suggesting that caliper-based measurements and 3D coordinates are equally well-suited for reconstructing genetic relationships, though our caliper-based data sets have the advantage to be many times larger.

Previous studies proposed that there are not only differences in neutrality between the four craniodental data types, but also differences among the variables within a given data type (10, 14, 17, 20). Our phenotypic variable undersampling procedure takes at least some of these considerations into account and we show that neutrality estimates for a given data type differ substantially when different subsets of variables are employed, further reinforcing previous claims. Future investigations should therefore explore additional arrangements of variables beyond the five tested here. For instance, considering only cranial data, combining all nonmetric variables, utilizing variables with the highest discriminatory power, or focusing on variables associated with previously identified functional and developmental modules (9, 17, 48). We propose that testing for neutrality in all possible combinations of cranial and dental metric and nonmetric variables, as recently employed for dental nonmetric trait data (10), is the most promising approach, rather than restricting analysis to predefined or hypothesized arrangements only.

In conclusion, our results serve as a reference for future craniodental research, confirming that most of the traditionally used data types can be used as proxies for neutral genomic data, although some are more useful than others. We do advise, however, to carefully review the use of dental metrics in the form of standard mesiodistal and buccolingual crown dimensions only, as they may not cover sufficient independent genomic variation, at least in comparison with other craniodental data types. Importantly, instead of using the different data types separately, we advise relying on a more holistic approach by combining them together, as this maximizes genotypic coverage over different loci resulting from primarily neutral evolution. Future work in combinatorics should focus on identifying specific subsets of mixed cranial and dental metric and nonmetric traits that are the most useful for tracking human neutral genetic variation.

Materials and methods

Matching population samples

Materials for this study comprise five different types of data: (i) SNPs, (ii) cranial metrics, (iii) dental metrics, (iv) cranial nonmetric traits, and (v) dental nonmetric traits. All data were taken from existing databases. We matched the different data types for 26 globally distributed modern human populations for which all five types of data were available (Fig. 1 and SI Appendix, Table S1). Populations were chosen for inclusion in this study based on three criteria: (i) availability of $n \ge 3$ unrelated individuals per genetic sample; (ii) availability of $n \ge 4$ individuals per phenotypic sample; and (iii) sample antiquity <2,000 years, to control for temporal bias. In instances where exact population matches between genotypic and phenotypic populations could not be achieved, a geographically proximate population with ethno-linguistic affinities was selected. In a few cases, closely related populations were pooled to maximize sample size. We note that the matched population samples in this study are unpaired; that is, all five types of data derive from different individuals. When possible, approximately equal numbers of adult males and females (determined osteologically) were sampled for the phenotypic data sets, to control for sexual dimorphism; however, we note that overall the phenotypic data sets are biased toward representing more males.

SNP data

SNP data were obtained from published databases, genotyped with the Affymetrix Human Origins Array (71-80). To correctly merge genotypes coming from different data sets, we ensured they were all related to the same Reference Sequence, the Genome Reference Consortium Human Build 37 (81) using, when needed, the LiftOver tool (82). To merge data from selected data sets, we used the plink-1.90 software (83). We filtered the data removing all transversions to avoid ambiguity in strand alignment (C/G or A/T), principal component analysis outliers, and first- and second-degree relative pairs. We selected only those SNPs that map to nonfunctional genomic regions and are therefore unlikely to be affected by natural selection. We applied two different filter levels for the amount of allowed missing data: first, to populations collected by Lazaridis et al. (74), Mallick et al. (76), Pickrell and Pritchard (78), and Skoglund et al. (80), we retained only individuals with 0% missing data; second, from the other published resources, we removed individuals with >10% of missing data. All filtering was performed using the plink-1.90 software (83). Finally, we converted the data set from PLINK file format into a genepop file using PGDSpider (84). The final preprocessed SNP data set comprised 857 individuals sharing 8,821 markers, with population sample representation varying from 3 to 176 individuals.

Cranial metric data

The cranial metric data were selected from a larger database collected by one of us (T.H.) (85). The data set consists of 37

measurements of the cranium recorded for each individual, in the form of linear dimensions, arcs, cords, and subtenses. All measurements were recorded following the procedures in Bräuer (70) using sliding and spreading calipers. Raw measurements were converted into scale-free shape variables by dividing each measurement by the geometric mean for all the measurements in each individual (86). This procedure removes gross size from the data in order to assess differences in the proportionate contribution of single variables to overall cranial size and adjusts for size differences between individuals that may result from sexual dimorphism. Because size-correction requires complete cases, missing values were imputed with the k-nearest neighbor (kNN) method (87). kNN searches the entire data set for k = 5 individuals most similar to the one with missing data and generates a mean to replace the missing value(s). Prior to imputation, individuals with more than half of the measurements missing were removed from the analysis. In this way, we ensured that <2.5% of the final data set requires imputation (down from 3.1%). Summary statistics of the kNN-imputed and size-corrected cranial metric data set are provided in Data Set S9. The final preprocessed cranial metric data set comprised 2,994 individuals, with population sample representation varying from 24 to 366 individuals.

Dental metric data

The dental metric data were selected from a larger database collected by one of us (T.H.) (88). The data set consists of mesiodistal and buccolingual crown diameters of all teeth recorded for each individual (up to a total of 28 metric variables, excluding third molars). Only right teeth were measured, but when a right tooth was missing, damaged, or affected by wear or pathology, the corresponding left antimere was measured. All measurements were recorded according to the procedures in Moorrees (89) and Hillson (90) using a digital sliding caliper accurate to 0.01 mm. We applied the same data preprocessing steps as for the cranial metric data. First, individuals missing more than half of the measurements were removed to ensure that <24.3% of the data set requires imputation (down from 57.7%). Second, missing values were imputed using the kNN algorithm (87). Third, raw measurements were converted into scale-free shape variables (86) to assess differences in the proportionate contribution of individual variables to overall tooth size and to adjust for size differences that may result from sexual dimorphism (40). Summary statistics of the kNN-imputed and size-corrected dental metric data set are reported in Data Set S10. The final preprocessed dental metric data set comprised 909 individuals, with population sample representation varying from 4 to 185 individuals.

Cranial nonmetric trait data

The cranial nonmetric trait data were selected from a larger database collected for the most part by one of us (T.H.) (91). The data set consists of 24 discrete observations of the cranium recorded for each individual and comprises data on sutural variation, supernumerary ossicles, hypostotic and hyperostotic traits, and vessel/nerve-related traits. The scoring procedures for each trait are described elsewhere [Hanihara et al. (91) and references therein]. Scoring followed the individual count method (92), where bilateral traits were counted only once per cranium, regardless of whether or not the trait appeared bilaterally. In cases where a trait was expressed asymmetrically, the side with the highest expression level was scored. Graded trait expression scores were collapsed into simplified binary dichotomies of absence or presence based on established breakpoints [Hanihara et al. (91) and references therein]. Sex differences were found in a few traits but none of the traits differed consistently between males and females in all sampled populations and we thus analyzed both sexes together, as it has been done in previous analyses of the same data set (91). Summary statistics of the cranial nonmetric trait data set are provided in Data Set S11. The final preprocessed cranial nonmetric trait data set comprised 4,623 individuals, with population sample representation varying from 26 to 533 individuals.

Dental nonmetric trait data

The dental nonmetric trait data were obtained from published resources (15, 68), whereby the majority of the samples were collected by C. G. Turner II, later augmented with samples collected by two of us (G.R.S. and J.D.I.; SI Appendix, supplementary text 1). The data set consists of 25 discrete observations of the dentition, including data on the number of cusps and roots, and the pattern of fissures, ridges, and grooves on tooth crowns. All data collectors used the Arizona State University Dental Anthropology System (ASUDAS) to record trait observations (68, 93). The ASUDAS comprises a reference set of dental casts illustrating expression levels for various traits alongside specific instructions that ensure a standardized scoring procedure, which minimizes observer error. ASUDAS traits are routinely collected on key teeth (usually the most mesial member of a tooth district) because these are considered the most stable members in terms of development and evolution (94). As in the cranial nonmetric trait data set, scoring followed the individual count method (92). Dental trait expression scores were collapsed into simplified binary dichotomies of absence or presence based on established breakpoints (15, 68). Dental traits of the ASUDAS have little or no sexual dimorphism, thus, it is a standard procedure to pool sexes (42, 46, 68, 94). Summary statistics of the dental nonmetric trait data set are provided in Data Set S12. The dental nonmetric trait data set comprised 2,986 individuals, with population sample representation varying from 28 to 450 individuals.

Estimating distances among populations

Pairwise neutral genetic distances among populations were computed from the SNP data using F_{ST} , defined as the fixation (F) index comparing the subset (s) genetic diversity within populations to the total ($_T$) genetic diversity of all sampled populations. We followed Weir and Cockerham's method of moments for diploid loci and calculated F_{ST} for each SNP individually, averaging F_{ST} over all loci (95). Under this model, populations of the same size are considered to have descended from a common ancestral population, which is assumed to be in Hardy–Weinberg equilibrium (Data Set S1).

Pairwise phenotypic distances were calculated from the craniodental data using Mahalanobis' D^2 distance, a model-free measure accounting for correlation among variables to avoid over-representing variation from variables that co-occur. The D^2 distance between two populations *i* and *j* is estimated as the difference between two vectors of variable averages (X_i and X_j), adjusted by a pooled within-population variance–covariance matrix (S) estimated over all populations in the analysis. For the cranial and dental metrics, we estimated D^2 following Mahalanobis (96), where X_i and X_j are calculated as geometric means, and S is calculated as a pooled Pearson variance–covariance matrix weighted by population sample sizes (Data Sets S2 and S3). For the cranial and dental nonmetric traits, we estimated D^2 following Nikita (97), where X_i and X_j are calculated as probit threshold values of trait frequencies, and S is calculated as a pooled Pearson correlation matrix weighted by the sample sizes for each pair of traits (Data Sets S4 and S5). When estimating D^2 for the combined craniodental data, we first computed D^2 independently for each of the four data types, and then combined the four D^2 matrices as a weighted average based on the numbers of variables (Data Set S6). Although this approach is valuable for handling unpaired samples and accounts for correlations within the four data sets, it does not account for correlations between them. However, in our case, it may still be appropriate since previous research demonstrated that the different data types are largely independent from each other, at least when comparing cranial metrics, dental metrics, and dental nonmetric traits (27), or cranial nonmetric and dental nonmetric traits (98). In addition to model-free D^2 distances, we also calculated model-bound P_{ST} distances, which incorporate relative estimates of effective population size (Ne; SI Appendix, Table S9) and average estimates of heritability (h^2 ; SI Appendix, supplementary text 2). Results obtained with P_{ST} show similarities to those using D^2 (SI Appendix, Table S10). However, due to the challenge of validating the parameter estimates N_e and h^2 , we opted to rely on D^2 in order to limit potential model bias.

Pairwise climatic distances among sampled population environments (C) were calculated as Euclidean distances based on five temperature-related variables obtained from a global climate database published in Hubbe et al. (9), using latitudes and longitudes approximated for each population sample (Data Set S7). As climate indicators for each population region, we used estimates of annual minimum temperature, annual maximum temperature, annual average temperature, maximum temperature of the warmest month, and minimum temperature of the coldest month, all measured in °C. These indicators are listed for each population sample in Data Set S13.

Pairwise geographic distances (G) were calculated as geodesic distances between population latitudes and longitudes (Data Set S8).

Correlation tests

We conducted Pearson correlation tests between the off-diagonal values in any two distance matrices to measure the linear association between phenotypic (D^2), genetic (F_{ST}), climate (C), and geographic (G) distances. We used partial Pearson correlation tests based on the residuals of a previous correlation and the offdiagonal values in a third matrix to evaluate the linear association between D^2 and F_{ST} , while controlling for either C or G. The resulting r coefficients are reported in SI Appendix, Table S2. To account for population sampling uncertainty in our partial correlation tests of D^2 , F_{ST} , and C, we calculated the r coefficients 1,000 times, each time leaving out a randomly selected population in the phenotypic and genomic data sets and a randomly selected individual in each remaining population. Additionally, to create equally sized numbers of variables across all phenotypic data sets, in each of the 1,000 iterations we randomly undersampled the number of variables down to 24, which corresponds to the number of variables in the cranial nonmetric trait data set, comprising the fewest variables among all phenotypic data sets being compared. Further, to account for stochastic variation inherent to a neutral model of evolution, in each of the 1,000 iterations we randomly undersampled the number of SNP loci down to the same number as there are phenotypic variables, namely, 24. To gauge the relative neutrality of the different phenotypic data types in a visual manner, we plotted the distributions of estimated r coefficients using violin plots. Statistical significance between pairs of distributions was evaluated with repeated-measures t-tests with the application of a Bonferroni correction for multiple testing (SI Appendix, Tables S3–S5).

Unless otherwise noted, all analyses were performed in R, version 4.2.2 (99). The data and R code are publicly accessible from the Zenodo repository at https://doi.org/10.5281/zenodo.8067443 . World map in Fig. 1 modified from https://commons.wikimedia. org/wiki/File:BlankMap-World6.svg (Public Domain).

Supplementary Material

Supplementary material is available at PNAS Nexus online.

Funding

This work was funded by the Senckenberg Society for Nature Research (Senckenberg Gesellschaft für Naturforschung) and by the German Research Foundation (Deutsche Forschungsgemeinschaft; DFG FOR 2237 and DFG INST 37/706-1 FUGG). K.H. is funded by the European Research Council (ERC AdG 101019659). H.R.C. is funded by the US National Science Foundation (Mid-scale RI-1 Project 2131940, EduceLab: Infrastructure for Next-Generation Heritage Science). Funding for data collection by J.D.I. came from the National Science Foundation (BCS-0840674, BNS-0104731, BNS-9013942), Arizona State University Research Development Program, and American Museum of Natural History. We acknowledge support from the Open Access Publishing Fund of the University of Tübingen.

Author Contributions

H.R. designed and performed the research; T.H., G.R.S., and J.D.I. collected data; H.R., S.P., and V.P. analyzed the data; H.R. and S.P. wrote the paper with contributions from all authors; H.R., S.G., and K.H. supervised the research.

Data availability

The data and code used for analyses are publicly accessible from the Zenodo repository at https://doi.org/10.5281/zenodo.8067443.

References

- 1 von Cramon-Taubadel N. 2014. Evolutionary insights into global patterns of human cranial diversity: population history, climatic and dietary effects. *J Anthropol Sci.* 92:43–77.
- 2 Weaver TD. 2018. Neutral theory and the evolution of human physical form: an introduction to models and applications. *J Anthropol* Sci. 96:7–26.
- 3 Schroeder L, Ackermann RR. 2023. Moving beyond the adaptationist paradigm for human evolution, and why it matters. J Hum Evol. 174:103296.
- 4 Savell KRR, Auerbach BM, Roseman CC. 2016. Constraint, natural selection, and the evolution of human body form. Proc Natl Acad Sci U S A. 113:9492–9497.
- 5 Betti L, von Cramon-Taubadel N, Lycett SJ. 2012. Human pelvis and long bones reveal differential preservation of ancient population history and migration out of Africa. Hum Biol. 84:139–152.
- 6 Betti L, Lycett SJ, von Cramon-Taubadel N, Pearson OM. 2015. Are human hands and feet affected by climate? A test of Allen's rule. Am J Phys Anthropol. 158:132–140.
- 7 von Cramon-Taubadel N. 2011. Global human mandibular variation reflects differences in agricultural and hunter-gatherer subsistence strategies. Proc Natl Acad Sci U S A. 108:19546–19551.

- 8 Noback ML, Harvati K. 2015. The contribution of subsistence to global human cranial variation. *J Hum Evol*. 80:34–50.
- 9 Hubbe M, Hanihara T, Harvati K. 2009. Climate signatures in the morphological differentiation of worldwide modern human populations. Anat Rec. 292:1720–1733.
- 10 Rathmann H, Reyes-Centeno H. 2020. Testing the utility of dental morphological trait combinations for inferring human neutral genetic variation. Proc Natl Acad Sci U S A. 117:10769–10777.
- 11 Katz DC, Grote MN, Weaver TD. 2017. Changes in human skull morphology across the agricultural transition are consistent with softer diets in preindustrial farming groups. *Proc Natl Acad* Sci U S A. 114:9050–9055.
- 12 Mitteroecker P, Grunstra NDS, Stansfield E, Waltenberger L, Fischer B. 2021. Did population differences in human pelvic form evolve by drift or selection? BMSAP. 33:10–25.
- 13 Ponce de León MS, et al. 2018. Human bony labyrinth is an indicator of population history and dispersal from Africa. Proc Natl Acad Sci U S A. 115:4128–4133.
- 14 Reyes-Centeno H, Ghirotto S, Harvati K. 2017. Genomic validation of the differential preservation of population history in modern human cranial anatomy. Am J Phys Anthropol. 162: 170–179.
- 15 Irish JD, Morez A, Girdland Flink L, Phillips ELW, Scott GR. 2020. Do dental nonmetric traits actually work as proxies for neutral genomic data? Some answers from continental- and global-level analyses. Am J Phys Anthropol. 172:347–375.
- 16 Roseman CC. 2004. Detecting interregionally diversifying natural selection on modern human cranial form by using matched molecular and morphometric data. Proc Natl Acad Sci U S A. 101: 12824–12829.
- 17 von Cramon-Taubadel N. 2011. The relative efficacy of functional and developmental cranial modules for reconstructing global human population history. Am J Phys Anthropol. 146:83–93.
- 18 Rathmann H, et al. 2017. Reconstructing human population history from dental phenotypes. Sci Rep. 7:12495.
- 19 Smith HF. 2009. Which cranial regions reflect molecular distances reliably in humans? Evidence from three-dimensional morphology. Am J Hum Biol. 21:36–47.
- 20 Harvati K, Weaver TD. 2006. Human cranial anatomy and the differential preservation of population history and climate signatures. Anat Rec. 288:1225–1233.
- 21 Brewster C, Meiklejohn C, von Cramon-Taubadel N, Pinhasi R. 2014. Craniometric analysis of European Upper Palaeolithic and Mesolithic samples supports discontinuity at the Last Glacial Maximum. Nat Commun. 5:4094.
- 22 Ousley S. 2016. Forensic classification and biodistance in the 21st century: the rise of learning machines. In: Pilloud MA, Hefner JT, editors. Biological distance analysis: forensic and bioarchaeological perspectives. London: Academic Press. p. 197–212.
- 23 Harvati K, Ackermann RR. 2022. Merging morphological and genetic evidence to assess hybridization in Western Eurasian late Pleistocene hominins. Nat Ecol Evol. 6:1573–1585.
- 24 Harvati K, et al. 2019. Apidima Cave fossils provide earliest evidence of *Homo sapiens* in Eurasia. Nature. 571:500–504.
- 25 Reyes-Centeno H, et al. 2014. Genomic and cranial phenotype data support multiple modern human dispersals from Africa and a southern route into Asia. Proc Natl Acad Sci U S A. 111: 7248–7253.
- 26 Harvati K. 2009. Into Eurasia: a geometric morphometric reassessment of the Upper Cave (Zhoukoudian) specimens. *J Hum* Evol. 57:751–762.
- 27 Rathmann H, Stoyanov R, Posamentir R. 2022. Comparing individuals buried in flexed and extended positions at the Greek

colony of Chersonesos (Crimea) using cranial metric, dental metric, and dental nonmetric traits. *Int J Osteoarchaeol*. 32:49–63.

- 28 Brasil MF, Monson TA, Schmitt CA, Hlusko LJ. 2020. A genotype: phenotype approach to testing taxonomic hypotheses in hominids. Sci Nat. 107:40.
- 29 Röding C, Zastrow J, Scherf H, Doukas C, Harvati K. 2021. Crown outline analysis of the hominin upper third molar from the Megalopolis Basin, Peloponnese, Greece. In: Reyes-Centeno H, Harvati K, editors. Ancient connections in Eurasia. Tübingen: Kerns Verlag. p. 13–36.
- 30 Slimak L, et al. 2022. Modern human incursion into Neanderthal territories 54,000 years ago at Mandrin, France. Sci Adv. 8: eabj9496.
- 31 Nikita E, Mattingly D, Lahr MM. 2012. Sahara: barrier or corridor? Nonmetric cranial traits and biological affinities of North African late Holocene populations. Am J Phys Anthropol. 147:280–292.
- 32 Torres-Rouff C, Knudson KJ, Hubbe M. 2013. Issues of affinity: exploring population structure in the Middle and Regional Developments Periods of San Pedro de Atacama, Chile. *Am J Phys Anthropol.* 152:370–382.
- 33 Hefner JT, Ousley SD. 2014. Statistical classification methods for estimating ancestry using morphoscopic traits. J Forensic Sci. 59: 883–890.
- 34 Irish JD, Bailey SE, Guatelli-Steinberg D, Delezene LK, Berger LR. 2018. Ancient teeth, phenetic affinities, and African hominins: another look at where Homo naledi fits in. J Hum Evol. 122:108–123.
- 35 Hlusko LJ, et al. 2018. Environmental selection during the last ice age on the mother-to-infant transmission of vitamin D and fatty acids through breast milk. Proc Natl Acad Sci U S A. 115: E4426–E4432.
- 36 Scott GR, et al. 2021. Peopling the Americas: not "out of Japan". PaleoAmerica. 7:309–332.
- 37 Scott GR, et al. 2018. rASUDAS: a new web-based application for estimating ancestry from tooth morphology. Forencic Anthropol. 1:18–31.
- 38 Martinón-Torres M, et al. 2007. Dental evidence on the hominin dispersals during the Pleistocene. Proc Natl Acad Sci U S A. 104: 13279–13282.
- 39 Stojanowski CM, Schillaci MA. 2006. Phenotypic approaches for understanding patterns of intracemetery biological variation. *Am J Phys Anthropol.* 131:49–88.
- 40 Hemphill B. 2016. Assessing odontometric variation among populations. In: Irish JD, Scott GR, editors. A companion to dental anthropology. Chichester: Wiley Blackwell. p. 311–336.
- 41 Stojanowski CM, Paul KS, Seidel AC, Duncan WN, Guatelli-Steinberg D. 2017. Heritability and genetic integration of tooth size in the South Carolina Gullah. *Am J Phys Anthropol.* 164:505–521.
- 42 Stojanowski CM, Paul KS, Seidel AC, Duncan WN, Guatelli-Steinberg D. 2018. Heritability and genetic integration of anterior tooth crown variants in the South Carolina Gullah. *Am J Phys Anthropol.* 167:124–143.
- 43 Stojanowski CM, Paul KS, Seidel AC, Duncan WN, Guatelli-Steinberg D. 2019. Quantitative genetic analyses of postcanine morphological crown variation. Am J Phys Anthropol. 168: 606–631.
- 44 Martínez-Abadías N, et al. 2009. Heritability of human cranial dimensions: comparing the evolvability of different cranial regions. J Anat. 214:19–35.
- 45 Cheverud JM, Buikstra JE. 1981. Quantitative genetics of skeletal nonmetric traits in the rhesus macaques on Cayo Santiago. II. Phenotypic, genetic, and environmental correlations between traits. Am J Phys Anthropol. 54:51–58.

- 46 Paul KS, Stojanowski CM, Hughes TE, Brook AH, Townsend GC. 2020. Patterns of heritability across the human diphyodont dental complex: crown morphology of Australian twins and families. Am J Phys Anthropol. 172:447–461.
- 47 Carson EA. 2006. Maximum-likelihood variance components analysis of heritabilities of cranial nonmetric traits. *Hum Biol.* 78:383–402.
- 48 Hlusko LJ, Sage RD, Mahaney MC. 2011. Modularity in the mammalian dentition: mice and monkeys share a common dental genetic architecture. J Exp Zool B Mol Dev Evol. 316:21–49.
- 49 Hefner JT, Spradley MK, Anderson B. 2014. Ancestry assessment using random forest modeling. J Forensic Sci. 59:583–589.
- 50 Lease LR, Sciulli PW. 2005. Discrimination between European-American and African-American children based on deciduous dental metrics and morphology. Am J Phys Anthropol. 126:56–60.
- 51 Herrera B, Hanihara T, Godde K. 2014. Comparability of multiple data types from the Bering Strait region: cranial and dental metrics and nonmetrics, mtDNA, and Y-chromosome DNA. Am J Phys Anthropol. 154:334–348.
- 52 Leinonen T, McCairns RJS, O'Hara RB, Merilä J. 2013. Q(ST)-F(ST) comparisons: evolutionary and ecological insights from genomic heterogeneity. Nat Rev Genet. 14:179–190.
- 53 González-José R, van der Molen S, González-Pérez E, Hernández M. 2004. Patterns of phenotypic covariation and correlation in modern humans as viewed from morphological integration. *Am J Phys Anthropol.* 123:69–77.
- 54 Evteev A, Santos P, Grosheva A, Reyes-Centeno H, Ghirotto S. 2021. Associations between human genetic and craniometric differentiation across North Eurasia: the role of geographic scale. In: Reyes-Centeno H, Harvati K, editors. Ancient connections in Eurasia. Tübingen: Kerns Verlag. p. 157–192.
- 55 Monson TA, Fecker D, Scherrer M. 2020. Neutral evolution of human enamel-dentine junction morphology. Proc Natl Acad Sci U S A. 117:26183–26189.
- 56 Evteev AA, Movsesian AA. 2016. Testing the association between human mid-facial morphology and climate using autosomal, mitochondrial, Y chromosomal polymorphisms and cranial nonmetrics. Am J Phys Anthropol. 159:517–522.
- 57 Movsesian AA, Vagner-Sapukhina EA. 2021. Infracranial versus cranial nonmetric traits and mtDNA data in the study of genetic divergence of human populations. Int J Osteoarchaeol. 31:809–819.
- 58 Hubbard AR, Guatelli-Steinberg D, Irish JD. 2015. Do nuclear DNA and dental nonmetric data produce similar reconstructions of regional population history? An example from modern coastal Kenya. Am J Phys Anthropol. 157:295–304.
- 59 Strauss A, Hubbe M. 2010. Craniometric similarities within and between human populations in comparison with neutral genetic data. *Hum Biol.* 82:315–330.
- 60 Whitlock MC. 2008. Evolutionary inference from QST. Mol Ecol. 17: 1885–1896.
- 61 Rogers AR, Harpending HC. 1983. Population structure and quantitative characters. Genetics. 105:985–1002.
- 62 Edge MD, Rosenberg NA. 2015. Implications of the apportionment of human genetic diversity for the apportionment of human phenotypic diversity. Stud Hist Philos Biol Biomed Sci. 52: 32–45.
- 63 Hallgrímsson B, et al. 2009. Deciphering the palimpsest: studying the relationship between morphological integration and pheno-typic covariation. Evol Biol. 36:355–376.
- 64 Mitteroecker P, Bookstein F. 2007. The conceptual and statistical relationship between modularity and morphological integration. Syst Biol. 56:818–836.

- 65 Klingenberg CP. 2008. Morphological integration and developmental modularity. *Annu Rev Ecol Evol Syst.* 39:115–132.
- 66 Witherspoon DJ, et al. 2007. Genetic similarities within and between human populations. *Genetics*. 176:351–359.
- 67 Hillson S, Fitzgerald C, Flinn H. 2005. Alternative dental measurements: proposals and relationships with other measurements. Am J Phys Anthropol. 126:413–426.
- 68 Scott GR, Irish JD. 2017. Human tooth crown and root morphology. Cambridge: Cambridge University Press.
- 69 Hauser G, De Stefano GF. 1989. Epigenetic variants of the human skull. Stuttgart: Schweizerbart'sche Verlagsbuchhandlung.
- 70 Bräuer G. 1988. Osteometrie: a. Kraniometrie. In: Knussmann R, editor. Anthropologie: Handbuch der vergleichenden Biologie des Menschen, Band I. Stuttgart: Gustav Fischer. p. 160–192.
- 71 Bergström A, et al. 2020. Insights into human genetic variation and population history from 929 diverse genomes. *Science*. 367: eaay5012.
- 72 Henn BM, et al. 2012. Genomic ancestry of North Africans supports back-to-Africa migrations. PLoS Genet. 8:e1002397.
- 73 Lao O, et al. 2008. Correlation between genetic and geographic structure in Europe. Curr Biol. 18:1241–1248.
- 74 Lazaridis I, et al. 2014. Ancient human genomes suggest three ancestral populations for present-day Europeans. Nature. 513: 409–413.
- 75 Lipson M, et al. 2018. Population turnover in remote oceania shortly after initial settlement. Curr Biol. 28:1157–1165.e7.
- 76 Mallick S, et al. 2016. The Simons Genome Diversity Project: 300 genomes from 142 diverse populations. Nature. 538:201–206.
- 77 Nelson MR, et al. 2008. The Population Reference Sample, POPRES: a resource for population, disease, and pharmacological genetics research. Am J Hum Genet. 83:347–358.
- 78 Pickrell JK, Pritchard JK. 2012. Inference of population splits and mixtures from genome-wide allele frequency data. PLoS Genet. 8: e1002967.
- 79 Qin P, Stoneking M. 2015. Denisovan ancestry in east Eurasian and native American populations. Mol Biol Evol. 32:2665–2674.
- 80 Skoglund P, et al. 2016. Genomic insights into the peopling of the Southwest Pacific. Nature. 538:510–513.
- 81 Church DM, et al. 2011. Modernizing reference genome assemblies. PLoS Biol. 9:e1001091.
- 82 Kuhn RM, Haussler D, Kent WJ. 2013. The UCSC genome browser and associated tools. Brief Bioinform. 14:144–161.
- 83 Purcell S, et al. 2007. PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Genet. 81:559–575.
- 84 Lischer HEL, Excoffier L. 2012. PGDSpider: an automated data conversion tool for connecting population genetics and genomics programs. Bioinformatics. 28:298–299.

- 85 Hanihara T. 1996. Comparison of craniofacial features of major human groups. Am J Phys Anthropol. 99:389–412.
- 86 Jungers WL, Falsetti AB, Wall CE. 1995. Shape, relative size, and size-adjustments in morphometrics. Am J Phys Anthropol. 38: 137–161.
- 87 Kenyhercz MW, Passalacqua NV. 2016. Missing data imputation methods and their performance with biodistance analyses. In: Pilloud MA, Hefner JT, editors. Biological distance analysis: forensic and bioarchaeological perspectives. London: Academic Press. p. 181–194.
- 88 Hanihara T, Ishida H. 2005. Metric dental variation of major human populations. Am J Phys Anthropol. 128:287–298.
- 89 Moorrees CFA. 1957. The Aleut dentition: a correlative study of dental characteristics in an Eskimoid people. Cambridge: Harvard University.
- 90 Hillson S. 1996. Dental anthropology. Cambridge: Cambridge University Press.
- 91 Hanihara T, Ishida H, Dodo Y. 2003. Characterization of biological diversity through analysis of discrete cranial traits. *Am J Phys Anthropol.* 121:241–251.
- 92 Turner CG II, Scott GR. 1977. Dentition of easter islanders. In: Dahlberg AA, Graber T, editors. Orofacial growth and development. The Hague: Mouton. p. 229–249.
- 93 Turner CG II, Nichol C, Scott GR. 1991. Scoring procedures for key morphological traits of the permanent dentition: The Arizona State University dental anthropology system. In: Kelley M, Larsen C, editors. Advances in dental anthropology. New York: Wiley-Liss. p. 13–32.
- 94 Scott GR, Turner CG II, Townsend GC, Martinón-Torres M. 2018. The anthropology of modern human teeth. Cambridge: Cambridge University Press.
- 95 Weir BS, Cockerham CC. 1984. Estimating F-statistics for the analysis of population structure. Evolution. 38:1358–1370.
- 96 Mahalanobis PC. 1936. On the generalized distance in statistics. Proc Nat Instit Sci Ind. 2:49–55.
- 97 Nikita E. 2015. A critical review of the mean measure of divergence and Mahalanobis distances using artificial data and new approaches to the estimation of biodistances employing nonmetric traits. Am J Phys Anthropol. 157:284–294.
- 98 Ricaut F-X, et al. 2010. Comparison between morphological and genetic data to estimate biological relationship: the case of the Egyin Gol necropolis (Mongolia). Am J Phys Anthropol. 143: 355–364.
- 99 R Core Team. R: a language and environment for statistical computing. Vienna (Austria): R Foundation for Statistical Computing. https://www.R-project.org/.